Author's response to reviews

Title: Population and antenatal-based HIV prevalence estimates in a high contracepting female population in rural South Africa

Authors:

Brian D Rice (briandrice@yahoo.com)  
Jorg Batzing-Feigenbaum (Baetzing-FeigenbaumJ@rki.de)  
Victoria Hosegood (Vicky.Hosegood@lshtm.ac.uk)  
Frank Tanser (tanserf@africacentre.ac.za)  
Caterina Hill (cats_hill@hotmail.com)  
Till Barnighausen (tbarnighausen@africacentre.ac.za)  
Kobus Herbst (kherbst@africacentre.ac.za)  
Tanya Welz (tanyawelz@btinternet.com)  
Marie-Louise Newell (mnewell@africacentre.ac.za)

Version: 5  Date: 16 July 2007

Author’s response to reviews:

Dear sir/madam - may i take this opportunity to thank you again for accepting our paper for publication.

As instructed the revised manuscript has had the excess figures / legends removed. No other changes were made to the manuscript.

All the best, Brian.